Author's response to reviews

Title: Germ-line mutations in Epidermal growth factor receptor (EGFR) are rare but may contribute to oncogenesis: a novel germ-line mutation in EGFR detected in a patient with lung adenocarcinoma

Authors:

Irene Centeno (centeno.ramos@gmail.com)
Pilar Blay (pilarblayalbors@gmail.com)
Iñigo Santamaria (isr@hca.es)
Aurora Astudillo (astudillo@hca.es)
Ana S Pitiot (anapitiot@uniovi.es)
Fernando G Osorio (fgosorio@degradome.uniovi.es)
Patricia Gonzalez-Arriaga (laboepic.uo@uniovi.es)
Fernando Iglesias (fernandoiglesiasl@hotmail.com)
Primitiva Menenedez (tiva@hca.es)
Adonina Tardon (atardon@uniovi.es)
Jose MP Freije (jmpf@uniovi.es)
Milagros Balbin (mbalbin@hca.es)

Version: 3 Date: 6 May 2011

Author's response to reviews:

Oviedo, May 5, 2011

Dear Editor:

Thank you very much for finally accepting our manuscript.

Please find enclosed a revised version of our manuscript entitled “Germ-line mutations in Epidermal growth factor receptor (EGFR) are rare but may contribute to oncogenesis: a novel germ-line mutation in EGFR detected in a patient with lung adenocarcinoma” by Centeno et al. with all the formatting changes made as suggested.

Major revisions:

Affiliations: We have written in full each affiliation.

Tables: We have created the tables with the tables tool and used the recommended template (simple1).

Supplementary table: We have renamed it as “Table 3”

Minor revisions:

References: They contain up to 30 authors’ names.
We hope that the article now conforms completely the journal style.

Thanking again for your consideration, sincerely,

Milagros Balbin, PhD
Laboratorio de Oncología Molecular
Instituto Universitario de Oncología del Principado de Asturias-IUOPA
Hospital Universitario Central de Asturias
33006 Oviedo Spain
(Manuscript ID 6785219994928199)